\begin{table}%t4
\caption{\label{phot}Input cluster parameters.}                  
%\centerline
 {         
\begin{tabular}{l c c c }     
\hline\hline       
Cluster   & $E(B-V)$ & $(m-M)_V$ & log Age \\
          & (mag)   & (mag)     & (dex) \\
\hline     	     
Cr~110\tablefootmark{a}
          & 0.54~$\pm$~0.03 & 13.37~$\pm$~0.38 & 9.23 $\pm $0.15 \\	  
NGC~2099 (M~37)\tablefootmark{b}
          & 0.27 $\pm$ 0.04 & 11.53 $\pm$ 0.19 & 8.61 $\pm$ 0.16 \\
NGC~2420\tablefootmark{c} 
          & 0.04 $\pm$ 0.03 & 11.88 $\pm$ 0.27 & 9.47 $\pm$ 0.17 \\
NGC~2682 (M~67)\tablefootmark{d}     
          & 0.04 $\pm$ 0.02 &  9.67 $\pm$ 0.11 & 9.64 $\pm$ 0.05 \\
NGC~7789\tablefootmark{e}  
          & 0.27 $\pm$ 0.04 & 12.23 $\pm$ 0.20 & 9.21 $\pm$ 0.12 \\
\hline                  
\end{tabular}}
\vspace*{-2mm}
\tablefoot{
\tablefoottext{a}{Averages of measurements by \citet{tsa71},
      \citet{daw98}, and \citet{bra06}.}
\tablefoottext{b} {Averages of measurements by \citet{west67a}, 
      \citet{jen75}, \citet{lynga87}, \citet{mer96}, \citet{tw97}, 
      \citet{kiss01}, \citet{kalirai01}, \citet{nila02}, \citet{gro03}, 
      \citet{bra06}, and \citet{har08}.}
   \tablefoottext{c} {Averages of measurements by \citet{mcc74},
      \citet{bar84}, \citet{van85}, \citet{lynga87}, \citet{tw90},
      \citet{car94}, \citet{dem94}, \citet{tw97}, \citet{tad02}, and
      \citet{gro03}.}   
\tablefoottext{d} {Averages of measurements by \citet{rac71},
      \citet{bar84},  \citet{van85}, \citet{nis87}, \citet{hob91},
      \citet{gil92},  \citet{dem92}, \citet{mey93}, \citet{mon93},
      \citet{car94}, \citet{din95}, \citet{fan96}, \citet{tw97},
      \citet{sar99}, \citet{tad02}, \citet{gro03}, \citet{van03},
      \citet{lau04}, and \citet{san04}.}
\tablefoottext{e} {Averages of measurements by \citet{bur58},
      \citet{arp62}, \citet{str70}, \citet{jen75}, \citet{jan77},
      \citet{cla79}, \citet{twa85}, \citet{lynga87}, \citet{maz88},
      \citet{friel93}, \citet{mar94}, \citet{jan94}, \citet{tw97},
      \citet{gim98a}, \citet{sar99}, \citet{val00}, \citet{tad02},
      and \citet{bar04}.}}
\end{table}